Author's response to reviews

Title: Evaluation of the middle cerebral artery occlusion techniques in the rat by in-vitro 3-dimensional micro- and nano computed tomography

Authors:

Alexander C Langheinrich (alexander.langheinrich@radiol.med.uni-giessen.de)
Mesut Yeniguen (Mesut.Yeniguen@neuro.med.uni-giessen.de)
Anne Ostendorf (Anne.M.Ostendorf@med.uni-giessen.de)
Simone Doenges (S.Marhoffer@gmx.de)
Marian Kampschulte (Marian.Kampschulte@radiol.med.uni-giessen.de)
Georg Bachmann (g.bachmann@kerckhoff-klinik.de)
Erwin Stolz (Erwin.Stolz@neuro.med.uni-giessen.de)
Tibo Gerriets (tibo.gerriets@neuro.med.uni-giessen.de)

Version: 4 Date: 17 May 2010

Author's response to reviews: see over
Dear Editor,

This is a re-submission of our revised manuscript entitled

“Evaluation of the Middle Cerebral Artery Occlusion Techniques in the Rat by in-vitro 3-Dimensional Micro- and Nano Computed Tomography”.

We answered concerns on a point-by-point basis and have indicated all changes made in the manuscript.

We hope that the present version of the manuscript is now acceptable for publication in your journal.

Yours sincerely,

Alexander C. Langheinrich

Major revisions (we require the author to make these changes)

Figures: please cite all figures in numerical order in the text. Figure 4 is currently cited in the wrong position.

AUTHORS: Due to the different methods used in the manuscript, the numerical order of the figures in the text can not be completely changed.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Qualifications: Please remove the authors' qualifications (e.g. PhD / MD) or job titles from the manuscript and submissions system

Authors: Authors qualifications have been removed from the manuscript.

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include
full stops/periods. For example, `The Title of My Prestigious Publication: A Scientific Study.¿ should be changed to `The title of my prestigious publication: a scientific study¿

AUTHORS: OK

Title page: remove word/table count.

AUTHORS: OK

Abstract: please change Purpose to Background and remove keywords from this section.

AUTHORS: OK

Introduction: rename Background.

AUTHORS: OK

Coloured text: please remove.

AUTHORS: OK

Page 18: please remove this blank page.

AUTHORS: OK

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

AUTHORS: OK